Author’s response to reviews

Title: Dental caries prevention strategies among children and adolescents with immigrant- or low socioeconomic backgrounds- do they work? A systematic review

Authors:

Marit Slåttelid Skeie (marit.skeie@odont.uib.no)

Kristin S Klock (kristin.klock@uib.no)

Version: 2 Date: 25 Sep 2017

Author’s response to reviews:

Bergen, Monday, Sept 25th, 2017

Editorial Board, BMC Oral Health

Thank you for the comments.

We hereby submit another revised manuscript of "Dental caries prevention strategies among children and adolescents with immigrant- or low socioeconomic backgrounds- do they work? A systematic review" for evaluation and possible publication in BMC Oral Health.

The manuscript has newly (Sept 20th) been received from the language editing service, “Write2Publish”, by Paul Riordan http://correction-home.fr/Write2Publish/Write2Publish_Home.html

We have followed the editorial requirements and the manuscript is in accordance with BioMed Central editorial policies and formatting guidelines.

Yours sincerely,